Author’s response to reviews

Title: Methodological quality of English language genetic guidelines on hereditary breast cancer screening and management: an evaluation using the AGREE instrument

Authors:

Benedetto Simone (benedetto.simone@yahoo.it)
Emma De Feo (emmadefeo@yahoo.it)
Nicola Nicolotti (nicola.nicolotti@gmail.com)
Walter Ricciardi (direzioneigiene@rm.unicatt.it)
Stefania Boccia (sboccia@rm.unicatt.it)

Version: 4 Date: 15 October 2012

Author’s response to reviews:

Dear Editors,

We have amended the manuscript as requested:
- the tables have been moved in the manuscript
- the shades of grey have been eliminated from table 2
- the reference to table 4 has been removed

Thank you!